Author's response to reviews

Title: PAIS: Paracetamol (Acetaminophen) In Stroke; protocol for a randomized, double blind clinical trial. [ISCRTN 74418480]

Authors:

Eric J van Breda (e.vanbreda@erasmusmc.nl)
Bart H van der Worp (h.b.vanderworp@neuro.azu.nl)
Maarten HA van Gemert (m.gemert@meandermc.nl)
Ale Algra (a.algra@umcutrecht.nl)
Jaap L Kappelle (j.kappelle@neuro.azu.nl)
Jan van Gijn (j.vangijn@neuro.azu.nl)
Peter J Koudstaal (p.j.koudstaal@erasmusmc.nl)
Diederik WJ Dippel (d.dippel@erasmusmc.nl)

Version: 2 Date: 14 June 2005

Author's response to reviews:

Dear Sir or Madam:

Herewith I'd like to resubmit the manuscript of the protocol of the PAIS study, named:
PAIS: paracetamol (acetaminophen) in stroke; protocol for a randomized, double blind clinical trial. [ISCRTN 74418480].

The following changes were made:
- The footnote was incorporated in the text.
- Corrections or completions of the references were made.
- The header conflicts of interest was specified more precise.

I hope that all commentary is successfully put into practice.

Yours faithfully,

Eric van Breda, on behalf of the pais investigators